Author's response to reviews

Title: Health disparities and advertising content of women's magazines: a cross-sectional study

Authors:

Susan C Duerksen (sduerksen@projects.sdsu.edu)
Amy Mikail (amikail@ucsd.edu)
Laura Tom (farbeyondthestars@hotmail.com)
Annie Patton (annepatton85@hotmail.com)
Janina Lopez (janinalopez@hotmail.com)
Xavier Amador (xamador@ucsd.edu)
Reynaldo Vargas (rvargas@ucsd.edu)
Maria Victorio (mvictori@ucsd.edu)
Brenda Kustin (bkustin@ucsd.edu)
Georgia R Sadler (gsadler@ucsd.edu)

Version: 10 Date: 5 August 2005

Author's response to reviews:

Dear Biomed Central Editorial Team,

The final formatting revisions have been made as requested. The footnotes are incorporated into the text and the references have been changed to list all author names.

Thank you.

Susan Duerksen